Author's response to reviews

Title: Anemia, diet and iron supplementation among children living with HIV: a prospective cohort study

Authors:

Anita Shet (anitashet@gmail.com)
P K Bhavani (bhavanipk@yahoo.com)
N Kumarasamy (kumarasamy@yrgcare.org)
Karthika Arumugam (karthikakumar@gmail.com)
S Poongulali (poongulali@yrgcare.org)
Suresh Elumalai (aesuresh2001@yahoo.co.in)
Soumya Swaminathan (soumyas@trcchennai.in)

Version: 2
Date: 5 June 2015

Author's response to reviews:

Thank you for the editorial changes requested.

Title Page:
The email ID is verified and corrected.

2. Line Numbering:
Line and page numbers have been inserted.

3. List of abbreviations: Included as directed.

4. Tables: Added to the main manuscript.